Author's response to reviews

Title: How do multi-stage, multi-arm trials compare to the traditional two-arm parallel group design - a reanalysis of 4 trials

Authors:

Friederike M-S Barthel (sophie@fm-sbarthel.de)
Mahesh KB Parmar (mp@ctu.mrc.ac.uk)
Patrick Royston (pr@ctu.mrc.ac.uk)

Version: 4 Date: 7 April 2009

Author's response to reviews:

Dear Production Team

The following changes were made to the manuscript:

Major revisions:
Affiliations - Each affiliation has been written in full
Corresponding author – this has been indicated and email address is given
Additional files - A section titled "Additional files" has been added at the end

Minor revisions
Section numbers have been removed.
Tables: vertical lines are no longer visible.

Your sincerely
Dr FM-S Barthel